Author's response to reviews

Title: Characteristics of attitude and recommendation of Oncologists toward exercise in South Korea: A cross sectional survey study

Authors:

Ji-Hye Park (hobbang1922@hotmail.com)
Minsuk Oh (andy00001111@hanmail.net)
Yong Jin Yoon (yoonyj@yonsei.ac.kr)
Chulwon Lee (Wakeford@yonsei.ac.kr)
Lee W. Jones (lee.w.jones@duke.edu)
Seung Il Kim (skim@yuhs.ac)
Nam Kyu Kim (namkyuk@yuhs.ac)
Justin Y. Jeon (jjeon@yonsei.ac.kr)

Version: 7 Date: 26 February 2015

Author's response to reviews: see over
Date: February 26th 2015  
Ms. Ref. No.: 1460220917129101 
Title: Characteristics of attitude and recommendation of Oncologists toward exercise in South Korea: A cross sectional survey study

To: Editor of BMC Cancer.

Please find attached via online submission, a copy of our revised manuscript entitled “Characteristics of attitude and recommendation of Oncologists toward exercise in South Korea: A cross sectional survey study” that we are resubmitting for further consideration for publication.

We thank the reviewers for his/her thorough review of the manuscript and for their helpful suggestions. We are very glad that we were able to improve the manuscript by implementing all of the suggestions and comments from the reviewer.

We feel the changes made to the manuscript have improved its clarity, and hope that it is now in a form acceptable for publication in BMC Cancer. Thank you for your continued kind consideration. We look forward to hearing from you.

Sincerely,

Justin Y. Jeon  
Professor, Department of Sport and Leisure Studies, Yonsei University